Author’s response to reviews

Title: Multiplicative Disadvantage of Being an Unmarried and Inadequately Insured Woman Living in Poverty with Colon Cancer: Historical Cohort Exploration in California

Authors:

Naomi R. Levitz (levitz@uwindsor.ca)
Sundus Haji-Jama (sundus.haji.jama@mail.utoronto.ca)
Tonya Munro (munro3@uwindsor.ca)
Kevin M. Gorey (gorey@uwindsor.ca)
Isaac N. Luginaah (iluginaa@uwo.ca)
Emma Bartfay (emma.bartfay@uoit.ca)
GuangYong Zou (gzou@robarts.ca)
Frances C. Wright (frances.wright@sunnybrook.ca)
Sindu M. Kanjeekal (sindu.kanjeekal@wrh.on.ca)
Caroline Hamm (caroline.hamm@wrh.on.ca)
Madhan K. Balagurusamy (madhan08@gmail.com)
Eric J. Holowaty (eric.holowaty@sympatico.ca)

Version: 5 Date: 14 January 2015

Author’s response to reviews:

Dear Editorial Staff Members:

We made the following minor editorial revisions to our manuscript.

1) Author titles were removed from the title page.
2) The most recent abstract was attached (manuscript and online form matched).
3) We revised the authors’ contributions section, now specifying all authors.

We believe that our manuscript is properly formatted. We hope that it will now meet with your acceptance for publication. Thank you for your editorial assistance.

Sincerely,

Kevin Gorey